Author's response to reviews

Title: The Spinal Stenosis Pedometer and Nutrition Lifestyle Intervention (SSPANLI) Randomized Controlled Trial Protocol

Authors:

Christy C Tomkins-Lane (clane@mtroyal.ca)
Lynne MZ Lafave (llafave@mtroyal.ca)
Jill A Parnell (jparnell@mtroyal.ca)
Ashok Krishnamurthy (akrishnamurthy@mtroyal.ca)
Jocelyn Rempel (jirempel@mtroyal.ca)
Luciana G Macedo (lmacedo@ualberta.ca)
Stephanie Moriartey (Stephanie.Moriartey@albertahealthservices.ca)
Kent J Stuber (kstuber@cmcc.ca)
Richard Hu (Richard.Hu@albertahealthservices.ca)
Philip P Wilson (phwilson@brocku.ca)
Yvette M Andreas (andreasyvette@gmail.com)

Version: 2  Date: 1 November 2013

Author's response to reviews:

It was requested that the text be edited to avoid overlap with the manuscript we submitted describing the pilot phase of this project. I have substantially edited the manuscript to ensure no overlap.

Cheers,
Christy Tomkins-Lane